Author's response to reviews

Title: Putting Prevention into Practice: Barriers and Facilitating Factors Perceived by General Practitioners: Qualitative Study

Authors:

Ulla Walter (walter.ulla@mh-hannover.de)
Uwe Flick (flick@asfh-berlin.de)
Anke Neuber (neuber@uni-kassel.de)
Claudia Fischer (cfischer@asfh-berlin.de)
Rugzan J Hussein (Hussein.Rugzan@MH-Hannover.DE)
Friedrich W Schwartz (schwartz.fw@mh-hannover.de)

Version: 5 Date: 31 August 2010

Author's response to reviews: see over
Dear Sir,

we would like to thank the reviewers and the Associate Editor for the careful and very constructive comments and recommendations. Please find enclosed our revised manuscript. Please see below how we responded to each of the points raised.

List of changes:
   I. Language corrections: All changes mentioned by the editor concerning language have been performed.
   II. Results: The results are now presented in the past tense.
   III. Explanation: The word “offers” has been clarified and it is highlighted in yellow.
   IV. Abstract: The abstract has been reviewed carefully.
   V. New information: One paragraph has been added on page 4 concerning GPs’ preventive advice in Germany. This is highlighted in yellow.
   VI. Updated information: On page 20, the current fee allocated for primary preventive measures is reported (4.83 Euro per insured per year in 2009). This is highlighted in yellow.
   VII. Language copyediting: Professional language corrections have been performed by English Manager Science Editing (http://www.sciencemanager.com/) as recommended by the Editor.

Language corrections followed the Editor’s recommendations and further changes were performed by the English Editing Service. Thus, the language corrections are not highlighted, as the whole article has changed.

Thank you very much for your time and effort!

Sincerely,

Ulla Walter
Prof. Dr. phil.

Attachments:
Original manuscript